Author's response to reviews

Title: Vascular basis of mucosal color

Authors:

Johannes Kleinheinz (jklein@uni-muenster.de)
Andre Buchter (Andre.Buechter@ukmuenster.de)
Thomas Fillies (Thomas.Fillies@ukmuenster.de)
Ulrich Joos (joos@uni-muenster.de)

Version: 2 Date: 1 June 2005

Author's response to reviews:

Dear Sir,

Enclosed you will find our revised manuscript. We have changed the text according to the advices of the reviewers, especially the language has been modified in the whole text.

Sincerely

Johannes Kleinheinz MD, DMD, PhD